Author's response to reviews

Title: Risk adjustment for inter-hospital comparison of primary cesarean section rates: need, validity and parsimony

Authors:

Maria Pia Fantini (mariapia.fantini@unibo.it)
Elisa Stivanello (elisa.stivanello2@unibo.it)
Brunella Frammartino (brunella.frammartino@unibo.it)
Anna Patrizia Barone (barone@asplazio.it)
Danilo Fusco (fusco@asplazio.it)
Laura Dallolio (laura.dallolio@alma.unibo.it)
Paolo Cacciari (paocac@tin.it)
Carlo Alberto Perucci (dirdip@asplazio.it)

Version: 4 Date: 4 August 2006

Author's response to reviews: see over
Author's response to reviews

Title: Risk adjustment for inter-hospital comparison of primary cesarean section rates: need, validity and parsimony

Authors:
  Maria Pia Fantini: mariapia.fantini@unibo.it
  Elisa Stivanello: elisa.stivanello2@unibo.it
  Brunella Frammartino: brunella.frammartino@unibo.it
  Anna Patrizia Barone: barone@asplazio.it
  Danilo Fusco: fusco@asplazio.it
  Laura Dallolio: laura.dallolio@alma.unibo.it
  Paolo Cacciari: paocac@tin.it
  Carlo Alberto Perucci: dirdip@asplazio.it

Version: 3  Date: 4/08/2006

Author's response to reviews: see over
Object:

MS: 1279588359634721
Risk adjustment for inter-hospital comparison of primary cesarean section rates: need, validity and parsimony . Dr Maria Pia Fantini et. al.

Thank you for consideration of our manuscript for publication in your journal. We have reviewed the above manuscript according to your comments.

Version: 3 Date: 04 Aug 2006

In this version we make the following changes:

Abstract: We added in the background section of the abstract an introduction to place the study in the context of the current knowledge in the field.

References: We cited references with numbers in bracket and added all the authors names.

Competing interests - we wrote ‘The authors declare that they have no competing interests’.

Authors' contributions - We added the the declaration: 'All authors read and approved the final manuscript'.

We are looking forward to receiving your reply.
Best regards, on behalf of all the authors.